The international platform of Ludwig-Maximilians University's Center for Economic Studies and the Ifo Institute





# The Subversive Nature of Inequality: Subjective Inequality Perceptions and Attitudes to Social Inequality

Andreas Kuhn

# **CESIFO WORKING PAPER NO. 6023 CATEGORY 2: PUBLIC CHOICE** JULY 2016

An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com from the RePEc website: www.RePEc.org • from the CESifo website: www.CESifo-group.org/wp

ISSN 2364-1428

**CESifo Center for Economic Studies & Ifo Institute** 

# The Subversive Nature of Inequality: Subjective Inequality Perceptions and Attitudes to Social Inequality

# Abstract

This paper shows that higher levels of perceived wage inequality are associated with a weaker (stronger) belief into meritocratic (non-meritocratic) principles as being important in determining individual wages. This finding is robust to the use of an instrumental-variable estimation strategy which takes the potential issue of reverse causality into account, and it is further corroborated using various complementary measures of individuals' perception of the chances and risks associated with an unequal distribution of economic resources, such as their perception of the chances of upward mobility. I finally show that those individuals perceiving a high level of wage inequality also tend to be more supportive of redistributive policies and progressive taxation, and that they tend to favor the political left, suggesting a feedback effect of inequality perceptions into the political-economic sphere. Taken together, these findings suggest that high levels of perceived wage inequality have the potential to undermine the legitimacy of market outcomes.

### JEL-Codes: D310, D630, J310.

Keywords: inequality perceptions, attitudes to social inequality, legitimacy of market outcomes, beliefs about the causes of economic success, support of redistribution, political preferences.

Andreas Kuhn Swiss Federal Institute for Vocational Education and Training Kirchlindachstrasse 79 Switzerland – 3052 Zollikofen andreas.kuhn@ehb-schweiz.ch

March 2016

I thank Hans Pitlik, Justina Fischer, Alois Stutzer, and several participants at the 2013 Annual Meeting of the European Public Choice Society for helpful comments and suggestions on an early draft of this paper, and Sally Gschwend-Fisher for proofreading the manuscript.

# 1 Introduction

Many OECD countries have seen significant increases in inequality in the past few decades, including the United States as well as several continental European countries (e.g. Atkinson, 2008). Besides their direct and obvious impact through the re-allocation of economic resources to and from different socio-economic groups, changes in a country's aggregate distribution might also have repercussions in the political-economic sphere if they influence individuals' beliefs about the legitimacy of market outcomes. Indeed, several theoretical papers (e.g. Alesina and Angeletos, 2005; Alesina *et al.*, 2012; Bénabou and Tirole, 2006) made a convincing argument that changes in the aggregate level of inequality might influence individuals' attitudes towards social inequality, such as their beliefs about the importance of non-meritocratic principles in determining individual wages, and that these changes in attitudes might eventually feed back into the effective distribution of economic resources because they might affect distributional parameters ultimately set by voters, such as marginal tax rates on income.

The goal of this paper is to directly test the hypothesis that inequality affects the legitimacy of market-based outcomes. The paper presents, against this broader background, empirical estimates of the effect of individuals' subjective perceptions of wage inequality in their country of residence on their attitudes to social inequality, focusing on individuals' subjective beliefs that meritocratic, or non-meritocratic, factors are important in determining wages. In contrast to most previous studies on the topic, however, this paper focuses on a broad array of attitudes the perceived level of inequality potentially affects, rather than only on a single attitudinal measure.<sup>1</sup> The second key feature of this study is that the empirical analysis is based on an individual-level measure of subjective inequality perceptions, rather than on some objective and highly aggregated measure of inequality. Using a unique measure of individual-level perceptions of wage inequality, I find that higher levels of perceived inequality are significantly and substantively associated with a stronger (weaker) belief in labor markets functioning according to non-meritocratic (meritocratic) principles. In this sense, a higher perceived level of wage inequality indeed appears to subvert the legitimacy of market outcomes. This result holds true

<sup>&</sup>lt;sup>1</sup>While previous evidence has mostly focused on the impact of inequality on subjective wellbeing, both Bjørnskov *et al.* (2013) and Schneider (2012) have argued that the perceived fairness of the wage distribution mediates the effect of inequality on wellbeing, suggesting that a focus on a broader set of outcomes might yield interesting insights into the overall attitudinal effects of inequality. Note, however, that neither of these studies has estimated the direct effect of inequality (perceptions) on attitudes.

when I account for potential reverse causality using an instrumental-variable approach, and I further corroborate it using a variety of complementary outcomes reflecting an individual's perceptions of the chances and risks associated with an unequal distribution of wages, his or her attitudes towards the role of government and progressive taxation, as well as his or her general political preferences. In line with the results for individuals' beliefs about the reasons for getting ahead, these additional estimates show that higher inequality perceptions lead to a stronger (weaker) awareness or salience of the risks (chances) potentially associated with inequality, as well as a stronger support of government intervention aiming to reduce existing inequalities, of progressive taxation, and of the political left. Finally, and importantly, I also find that those individuals with higher inequality perceptions are less satisfied with their own wages.<sup>2</sup>

The key conceptual challenge in addressing these questions empirically is the need for an individual-level measure of inequality perceptions. Simply relying on a more or less easily available objective, yet also highly aggregated, measure of inequality would ignore the fact that people appear to hold widely different, and often heavily biased perceptions of the overall distribution of economic resources (e.g. Chambers *et al.*, 2014; Cruces *et al.*, 2013; Kuhn, 2011, 2015; Norton and Ariely, 2011; Osberg and Smeeding, 2006).<sup>3</sup> If we acknowledge that individuals have different, and presumably biased, perceptions of wage inequality, the use of an objective measure of inequality in the empirical analysis would simply fail to reflect heterogeneity of this kind, potentially implying inconsistencies in the estimated effect of inequality on attitudes.<sup>4</sup> In fact, the available empirical evidence shows that individuals' perceptions of wage levels and/or inequality are, in part, directly related to socio-economic factors (e.g. Kuhn, 2011; Schneider, 2012), suggesting that part of the observed variation in individual-level (mis-)perceptions of inequality is indeed of a systematic nature (implying a potential bias when this source of variation is ignored when estimating the impact of inequality on attitudes). To take these issues into

<sup>&</sup>lt;sup>2</sup>This last finding is insofar especially important as it allows me to directly relate my findings to the existing empirical evidence on the subject which has, as discussed in more detail below, up until now almost exclusively focused on the association between inequality and individuals' subjective wellbeing (Senik, 2005, provides a survey of this literature).

<sup>&</sup>lt;sup>3</sup>Other studies have focused on individual perceptions of alternative economic phenomena, such as tax rates (Gemmell *et al.*, 2004) or corruption (Olken, 2009).

<sup>&</sup>lt;sup>4</sup>There is, of course, also a more practical reason for favoring individual-level inequality perceptions over aggregate-level inequality measures. Because there is much more variation in individual-level inequality perceptions, it will presumably be easier to get reasonably precise estimates when using inequality perceptions rather than some objective measure of inequality.

account, the empirical analysis of this paper will therefore be based on a simple yet intuitive empirical measure of individual-level inequality perceptions, following the procedure proposed by and described in more detail in Kuhn (2011).

In contrast to the approach chosen in this paper, however, most previous papers studying the impact of inequality on individuals' attitudes used some objective, rather than subjective, measure of inequality as their key regressor – assuming, at least implicitly, that people correctly perceive the prevailing level of inequality (see Clark and D'Ambrosio, 2015, for a comprehensive survey). While earlier studies focused on highly aggregated measures of inequality (Verme, 2011, provides a survey of studies focusing on the association between country-level inequality measures and individual-level measures of subjective wellbeing), more recent studies have shifted their focus towards more disaggregated measures of inequality, arguing that people do not care as much about the aggregate distribution of market wages as they care about the comparison to smaller and much more specific reference groups, such as their coworkers, their friends, and/or their neighbors.<sup>5</sup> The latter group of studies does allow for variation in inequality perceptions across individuals, but only because of different comparison groups, not because of biased perceptions of wages or wage differentials per se.

Remarkably, these studies consistently tend to find a negative association between inequality and subjective wellbeing, independent of the exact definition of the comparison group.<sup>6</sup> For example, Clark and Oswald (1996) show in a sample of British workers that both job and wage satisfaction are significantly and negatively associated with an individual's comparison income, given by the predicted wage from a conventional wage regression, while holding the absolute level of income constant. Ferrer-i-Carbonell (2005) reports a similar result for Germany. She defines individuals with similar education, age, and location of residence as the comparison group and finds that well-being increases the larger an individual's income is with respect to that of the comparison group while holding the absolute level of income constant. Luttmer (2005), in contrast, matches individual-level survey data to information on local average household incomes from representative survey data from the US, using a strictly spatially defined

<sup>&</sup>lt;sup>5</sup>This lines up well with Clark and Senik (2010), who find that most individuals, if asked directly, confirm that such comparisons are at least somewhat important for them. Moreover, individuals most often compare themselves with their colleagues (i.e. their coworkers).

 $<sup>^{6}</sup>$ Only few papers have directly focused on behavioral, instead of attitudinal, effects of inequality, such as on individual work effort. In contrast to the (much larger) literature focusing on subjective wellbeing, the findings are considerably more mixed (e.g. Charness and Kuhn, 2007; Clark *et al.*, 2010; Cornelissen *et al.*, 2011).

comparison group. He finds that, even after controlling for an individual's own income, selfreported happiness is lower the higher average earnings are in an individual's neighborhood. Clark *et al.* (2009), using a similar approach and using data from Denmark, show that subjective well-being depends on one's relative rank within narrowly defined neighborhoods. Similarly, Schwarze and Härpfer (2007) report a negative relation between pre tax-and-transfer inequality and life satisfaction across regions in West Germany, while Winkelmann and Winkelmann (2010) report a negative association between aggregate-level inequality measures and economic wellbeing across municipalities in Switzerland as well.

The assumption that people have unbiased perceptions of the prevailing level of inequality has, however, been relaxed in two more recent studies. Both studies manipulate individuals' information about wage differentials in a field experiment. Both studies provide indirect evidence of people potentially having misperceptions of the true level of inequality, since a minor information treatment is able to influence individuals' attitudes. In the first of these studies, Card *et al.* (2012) randomize the access to information about coworkers' wages in a sample of university employees, providing arguably more compelling evidence on the causal effect of coworkers' wages on one's job satisfaction than most studies mentioned so far. Consistent with the evidence from the majority of observational studies, however, they also find that comparison income has a strong negative effect on individual job satisfaction. Finally, in another randomized field experiment, Kuziemko *et al.* (2015) manipulate individuals' information about income inequality. They find that providing individuals with this information has large effects on whether inequality is viewed as a pressing problem, but only small effects on individuals' support for governmental intervention with regard to the distribution of income.

Finally, yet on a somewhat more general level, this paper also adds to the small number of empirical studies documenting that economic events or circumstances influence individuals' attitudes and beliefs. In one of these studies, for example, Di Tella *et al.* (2007) show that granting property rights to squatters strongly influences their market beliefs. Similarly, Giuliano and Spilimbergo (2014) show that the macroeconomic context at the time of youth (i.e. growing up during a boom versus during a recession) influences individuals' beliefs about the importance of own effort and luck in getting ahead, their demand for redistribution, as well as their trust in public institutions. In line with the results from these studies, this paper shows that attitudes towards inequality are, in part, shaped by the prevailing level of inequality, and thus the paper also adds to our general understanding of how attitudes and preferences are formed and how they evolve over time.

The remainder of this paper proceeds as follows. The next section presents the data sources and discusses the selection of the sample as well as the construction of the key variables, i.e. attitudes to social inequality and individual-level inequality perceptions. Section 3 presents a few descriptives for the key variables. Section 4 discusses the econometric framework, while section 5 presents and discusses the resulting estimates. Section 6 concludes.

# 2 Data and Measures

### 2.1 Data Source

The empirical analysis fully relies on data from the "Social Inequality" cumulation of the years 1987, 1992, 1999, and 2009 (ISSP Research Group, 2014a,b). These data are a collection of several national-level surveys focusing specifically on attitudes to social inequality, administered and made available by the International Social Survey Programme (ISSP).<sup>7</sup> They represent one of the best available sources of quantitative data to study individuals' perceptions of inequality and their attitudes to social inequality across countries and over time.

One of the key advantages of these data is that they cover a large number of observations from various countries and that they stretch over a relatively long period of time, for at least a subset of the participating countries (appendix table A.1 shows the number of individual observations by country and year). Another attractive feature of the data is that they contain a large number of items related to individuals' attitudes to social inequality, including, for example, individuals' beliefs about the causes of economic success or their stated support of progressive taxation. On the other hand, however, there have been many changes to the survey design over time, with the practical consequence that the number of items available in all four rounds of the survey is rather limited (more on this below; see sections 3.1 and 4.2).

<sup>&</sup>lt;sup>7</sup>More information is available from the organization's website (www.issp.org). The data are available to researchers from the GESIS data archive (www.gesis.org/en/issp/home).

# 2.2 Sample Selection

The sample used in the main part of the empirical analyses is determined by the availability of the dependent variable and the key regressor (i.e. individual-level inequality perceptions). As noted above, relatively few of the survey items are available in identical form in all four years of the survey. This implies that there are potentially large differences in the effective sample size for the different outcome measures, irrespective of any additional nonresponse issues. Fortunately, however, the series of items used to construct the key regressor are available in all four years (though not necessarily for all countries) in principle, but there have been a few changes over time to that set of items as well (cf. footnote 9 below).

### 2.3 Attitudes to Social Inequality

The following paragraphs discuss the various outcome measures available from the ISSP surveys (see also table 1, which contains the full list of survey items used as outcomes, and which is discussed in more detail in section 3.1 below).

#### Beliefs About the Causes of Economic Success

The outcome measures of primary interest are several attitudinal questions from the ISSP surveys which explicitly focus on individuals' beliefs about the causes of economic success. Overall, there are twenty-six different items dealing with individuals' beliefs on this issue, and they can be further grouped into three distinct sets of items describing different dimensions of individuals' beliefs about the causes of economic success, namely: (i) beliefs about the reasons for getting ahead, (ii) beliefs about the incentivizing effect of inequality, and (iii) pay norms. Note that the first two sets of items relate to individuals' perception of the wage-setting process, while the third set of items asks individuals about their normative views regarding the factors that should ideally be important in determining wages.

The first group of items asked individuals about the reasons, from their subjective point of view, for getting ahead in their country of residence. For example, one of the items in this group asked respondents whether they thought that coming from a wealthy family was important for getting ahead (here, as for most other items, respondents could indicate their agreement with the statement on a five-point scale running from one to five; see table 1 for details). Because

this group of items contains variables describing both meritocratic (e.g. education and effort) as well as non-meritocratic (e.g. knowing the right people) mechanisms, an even more nuanced analysis of this specific dimension of attitudes to social inequality will be possible.

Then there are, secondly, several questions asking individuals' whether they believe that inequality has, at either the individual or the aggregate level, an incentivizing effect on economic actors. For example, respondents were asked whether they thought that wage differentials between people with different educational attainment are necessary as an individual incentive for choosing a higher educational attainment in the first place, or whether they thought that large differences in pay are necessary for a country's prosperity.

Finally, a third set of items asked individuals to state how wages should ideally be determined from their point of view. In other words, individuals were asked about their subjective pay norms (somewhat mirroring the first group of items, which asked them about the actual mechanisms determining pay). Here, for example, people were asked to state whether they thought that individual pay should be determined according to the attained level of education, or whether they would prefer compensation to depend on whether someone has to support children. As for individuals' beliefs about the reasons for getting ahead, it is also possible to distinguish between pay norms related to either meritocratic or non-meritocratic principles.

### The Perception of Chances and Risks Associated with Inequality

A second broader subject the survey covers, composed of a total of eleven distinct items, describes individuals' perceptions of both chances and risks potentially associated, at least in respondents' perceptions, with economic resources being distributed more or less unequally. The main hypothesis here is individuals who perceive wage inequality to be higher will tend to perceive chances to be less obvious and risks to be more relevant than individuals who perceive inequality to be low.

A first item of interest in this context asked individuals whether they thought that they "stand a good chance of improving their standard of living", which might be thought of as a subjective measure of perceived upward mobility. One might hypothesize that those individuals who perceive a high level of inequality are, ceteris paribus, less likely to agree with this statement if they, as argued above, perceive that non-meritocratic factors drive wage differentials. That is, assuming that people perceive access to higher wages to be mainly driven by non-meritocratic mechanisms, it appears likely that those who perceive a high level of wage inequality are less likely to believe in their own upward-mobility.

Another potentially interesting outcome in this context is individuals' satisfaction with their own wages. Again, if individuals believe that wages are (co-)determined by factors beyond an individual's control, such as coming from a wealthy family or having well-educated parents, then we would probably also expect them to be less satisfied with their own wages (i.e. an individual's own compensation from work likely appears unfair, relative to those with higher wages, if higher wages are mostly achieved through factors not related to one's own ability or effort).<sup>8</sup>

A third item of interest describes individuals' normative assessment of the overall level of inequality (as they personally perceive it). In this regard, respondents were directly asked to state if they thought that "income differences are too large" in their country of residence. Note that the item does not ask about the perception of the absolute level of inequality, but about the perceived level of inequality relative to some legitimate level.

Fourth and finally, individuals were also asked to evaluate whether any broader conflicts exist in society from their point of view, such as between the poor and the rich, or between unemployed and employed workers. The hypothesis here is that a higher perceived level of wage inequality goes hand in hand with people being more likely to perceive more general conflicts between different socio-economic groups in their society.

### The Role of Government and Support of Progressive Taxation

A further set of interesting items relates to the proper role of government towards redistribution and the provision of goods and services and to individuals' explicit support of progressive taxation. Again, if higher inequality perceptions are associated with a weaker belief in market justice, one would presumably also expect that higher inequality perceptions go hand in hand with a stronger support for government intervention and progressive taxation – to counteract the perceived malfunctioning of markets in the allocation of economic resources.

In the survey, individuals were asked with regard to the role of government to evaluate

<sup>&</sup>lt;sup>8</sup>Also, as noted in the introduction, almost all of the previous empirical papers on the subject have focused exclusively on the association between inequality and some measure of wage or general satisfaction; this outcome thus also allows me to compare and relate my own results to that specific strand of the literature.

whether they thought that the government is responsible for offering access to several goods and services such as providing a decent living standard for the unemployed. Regarding progressive taxation, individuals had to evaluate whether people with high incomes should pay higher taxes, or whether they thought that taxes for people with high income are high enough or not.

#### **Political Preferences and Political Participation**

Finally, the ISSP surveys also contain a few items on respondents' general political preferences; that is, their preferences towards either the political left or the political right, as well as their participation at the most recent national-level elections. I aggregate the three available items on political identification, as all three items ask about individuals' party preferences, and I use the remaining item on whether someone had voted at the last election in one's country as a rough measure of an individual's political participation.

### 2.4 Inequality Perceptions and Distributional Norms

### **Inequality Perceptions**

One of the most distinguishing features of the ISSP data on social inequality is a series of questions about the level of compensation for people working in various specific occupations, such as an unskilled worker in a factory or a doctor in general practice. For each of these occupations, respondents in the survey were asked to estimate both the actual level of earnings as well as the level of earnings that they would consider as legitimate (actual and ethical wage estimates, for short, in the following).<sup>9</sup>

Noting that an individual's set of actual wage estimates roughly describes the wage differentials he or she perceives to exist, these wage estimates can be used to construct a simple yet highly intuitive measure of inequality perceptions at the level of the individual, denoted by  $G_i^{\text{actual}}$  in what follows (see Kuhn, 2011, 2015, for additional details concerning the construction and interpretation of this measure). Put simply, though,  $G_i^{\text{actual}}$  is probably best thought of as

<sup>&</sup>lt;sup>9</sup>The list of occupations for which these individual wage estimates were elicited changed considerably over the different rounds of the survey. Fortunately, however, four specific occupations were part of this list in all four rounds of the survey; namely: (i) an unskilled worker in a factory, (ii) a doctor in general practice, (iii) a cabinet minister in the national government, and (iv) the chairman of a large national company. In the construction of individual-level inequality perceptions, I only use wage estimates for these four occupations to make the measure as comparable as possible over time.

aggregating the information available in the set of actual wage estimates for any given individual into a single number and, in this sense, as roughly approximating an individual's perceived overall distribution of market wages. In fact,  $G_i^{\text{actual}}$  is actually constructed in such a way that it has the same interpretation as a conventional Gini coefficient – with the key difference being that  $G_i^{\text{actual}}$  does not describe the "true" (i.e. objectively measured) distribution of wages but the distribution of wages as perceived by individual *i*. Consequently, note that there is variation in the Gini coefficient because of its subjective nature – in contrast to the conventional, objective measurement of inequality where there is but one Gini coefficient for a given country and a given point in time – as long as individuals have different perceptions of what people in different occupations actually earn. This turns out to be true empirically, as mentioned in the introduction and as shortly discussed in section 3.2 below.

### **Distributional Norms**

An analogous argument suggests that individuals' estimates of occupational wages that they judge as legitimate can be used to construct an individual-level measure of the legitimate level of inequality, denoted by  $G_i^{\text{ethical}}$  in what follows. Similar to the individual inequality perceptions above,  $G_i^{\text{ethical}}$  may be best understood as a rough summary measure of the wage distribution that a given individual *i* judges as fair, based on his or her set of ethical wage estimates for people working in different occupations, and thus  $G_i^{\text{ethical}}$  roughly reflects an individual's distributional norms. As the measure is constructed in the same way as individuals' inequality perceptions, it can also be interpreted as a Gini coefficient. In contrast to  $G_i^{\text{actual}}$  however,  $G_i^{\text{ethical}}$  describes a purely hypothetical situation, and thus there is no corresponding aggregate statistic as in the case of inequality perceptions.

Practically, it might be important to control for distributional norms when estimating the effect of inequality perceptions on attitudes towards social inequality because they are likely to have an influence individuals' attitudes to social inequality per se, and because previous empirical studies have shown that inequality perceptions and distributional norms are strongly correlated with each other (Kuhn, 2011).

# **3** Descriptives

### 3.1 Attitudes to Social Inequality

Table 1 presents a few sample descriptives for the full set of outcome measures eventually considered in the empirical analysis below. Whenever possible, items were recoded in such a way that larger values indicate a stronger degree of agreement with the underlying question or statement from the survey. Note that a common issue across many of the items listed in table 1 is that they do not appear in all four rounds of the survey (section 4.2 below will discuss how I deal with this issue empirically).

### Table 1

Panels (a) to (c) show descriptives for all the items describing an individual's beliefs about the causes of economic success, the functional role an unequal distribution of wages eventually plays, and about different pay norms. All related items are measured on a scale running from 1 to 5. Interestingly, the mean values from panels (a) to (c) suggest quite unambiguously that people, on average, have quite a strong belief in the proper functioning of labor markets, as they tend to perceive meritocratic mechanisms to be considerably more important than nonmeritocratic ones. Specifically, mean item values for meritocratic principles are close to about 4 throughout, while averages for items asking about the importance of non-meritocratic principles are mostly below 3 (implying that these factors, on average, are not regarded as important in determining one's pay).

Consistent with this pattern, panel (d) shows that individuals also tend to believe that they have a good chance of improving their standard of living (mean item value of about 3, also on a scale taking on values between 1 and 5). Nonetheless, most people appear to be somewhat dissatisfied with their own wages, according to the mean values of the two items related to wage satisfaction, both equal to about 2.3 points (again on a scale running from 1 to 5). Next, panel (e) shows that an overwhelming majority of the sample perceives the income differences in their country of residence to be too large (mean item value of more than 4 on a scale from 1 to 5). This also appears to be reflected in the mean values for the items relating to the perception of conflicts within society, as the items most clearly related to groups with large income differentials appear to receive the strongest agreement. Generally, it appears that many people perceive a relatively high level of conflict in their societies (note that the scale runs from 1 to 4 only for the items in panel (e); the mean values of these items are therefore not directly comparable to the averages of the other items).

Descriptives for individuals' beliefs about the proper role of government, as well as their support for progressive taxation, are shown in panel (f). It is evident from the descriptives that a majority of the sample believes that the government should play a decisive role in the provision of a number of services, including the financing of education and the provision of social security (most items averaging values between about 3.6 and 4), and that most individuals are in support of progressive taxes as well (average agreement equal to 3.3 and 4, respectively).

Finally, panel (g) shows that the average respondent in the pooled ISSP sample weakly tends to identify with the political left (here, the scale again runs from 1 to 5, with higher values denoting that an individual leans towards the political right). Finally, about 79% of the respondents state that they participated at the last elections. Below, I will use this item as a rough measure of an individual's level of political participation.

### 3.2 Inequality Perceptions

Panel (a) of table 2 presents descriptives for subjective inequality perceptions. In the overall sample, inequality perceptions average 0.56 Gini points with a standard deviation of about 0.218 (remember that inequality perceptions can, in principle, be interpreted the same way as a conventional Gini coefficient).

#### Table 2

The table also shows that very few individuals (less than one percent of the overall sample) perceive no wage differentials across the different occupations at all, resulting in  $G_i^{\text{actual}}$  being exactly equal to zero.<sup>10</sup> More important for the empirical analysis, however, note that there is huge variation in individual perceptions of wage inequality. This quite clearly shows that individuals really do have widely different perceptions of wage inequality, consistent with the

<sup>&</sup>lt;sup>10</sup>Somewhat counterintuitively, the table also shows that there are a few observations with negative inequality perceptions. This results if an individual reverses the rank ordering of the occupations (with respect to their wages) for which subjective wage estimates are available, and which are used to construct individual-level inequality perceptions. Kuhn (2011) provides a more detailed discussion of this issue.

literature mentioned in the introduction. This in turn implies that there might be large discrepancies between the effective level of inequality and the level of inequality individuals perceive.<sup>11</sup> At the same time, the high intraclass correlation (about 0.395) suggests that inequality perceptions are quite strongly correlated with each other within country×year cells. Thus, there appear to be large differences in country averages of individual-level inequality perceptions, reflecting not only differences in the true level of inequality, but also differences in, for example, the way and intensity the issue of inequality is treated in the media.

## 3.3 Other Individual-level Controls

Panel (b) of table 2 shows descriptives for the other three individual-level controls used in the analysis below. The descriptives of age and gender are as expected, with a mean age of about 45 years and slightly more women than men in the sample. More interestingly, note how small the intraclass correlation for these two variables is – compared to inequality perceptions.

Finally, descriptives for distributional norms reveal, not surprisingly, that the legitimate level of wage inequality is considerably smaller than the perceived level. Also, the corresponding intraclass correlation shows that distributional norms are less strongly correlated within country×survey-year than inequality perceptions.

# 4 Empirical Framework

### 4.1 Baseline Estimates

To quantify the effect of inequality perceptions on individuals' attitudes to social inequality, I will estimate the parameters of a series of regression models that all share the following basic form:

$$y_{it} = \alpha + \beta G_{it}^{\text{actual}} + \gamma x_{it} + \psi_{j[i]t} + \epsilon_{it}, \qquad (1)$$

with  $y_{it}$  denoting attitudes to social inequality of individual *i* living in country *j* and having participated in the survey in year *t*, and with  $G_{it}^{\text{actual}}$  denoting an individual's perceived level of

<sup>&</sup>lt;sup>11</sup>This, of course, also implies that there may be research contexts where simply working with objective data on inequality, instead of inequality perceptions, might be misleading.

wage inequality in his or her country of residence.  $\beta$  is the parameter of primary interest because it quantifies, under appropriate specifications of the model, the partial effect of inequality perceptions on individuals' attitudes to social inequality (oviously, I expect  $\beta > 0$ ).

Furthermore, all of the specifications shown below include a full set of country×surveyyear fixed effects, denoted by  $\psi_{j[i]t}$  in the equation above. The fixed effects will pick up any systematic differences across countries and years, even if not directly observable in the data at hand. Most importantly, perhaps, note that the inclusion of these fixed effects will net out any existing differences across countries and time in the objective, aggregate level of wage inequality. Moreover, most of the regression specifications shown below will add a few individual-level control variables, denoted by  $x_{it}$  in equation (1). Because most of the available controls at the individual level are potentially endogenous in the current context, however, I will only include few such controls; namely: age, gender, and distributional norms.<sup>12</sup> A second, more practical but no less important issue is the fact that there is a considerable amount of missing information with respect to many of these potential controls (due to both nonresponse and, again, to the fact that many potentially interesting controls do not appear consistently in all four rounds of the survey). Thus, the inclusion of additional controls would also lead to an often much smaller, and potentially very selective, subsample available for the empirical analysis.<sup>13</sup>

That said, there is of course a similar concern with inequality perceptions and distributional norms being endogenous as well. This concern is more difficult to deal with empirically, and it is therefore held back until section 4.3 below.

# 4.2 Stacking the Data to Estimate Average Effects within Groups of Similar Items

With a total of forty-nine distinct survey items, there are almost too many potentially interesting outcome variables. It is therefore desirable to have a succinct way to estimate some kind

 $<sup>^{12}</sup>$ In other words, most potential control variables at the individual level are suspect of being influenced by inequality perceptions, implying that specifications including these variables as controls would potentially bias the estimate of parameter  $\beta$  (i.e. these variables probably constitute "bad controls" in the language of Angrist and Pischke, 2008). For example, one can imagine that individuals who have been socialized to have a strong belief in non-meritocratic principles might choose a different educational track or occupation than those who believe that these mechanisms are not very important.

<sup>&</sup>lt;sup>13</sup>Nonetheless, appendix table A.3 shows a few selected estimates based on a larger set of individual-level controls. Reassuringly, the comparison shows that the inclusion of additional controls does not have a large impact on the interesting parameter estimates.

of average effects across items that focus on the same broader conceptual issue.

For this reason, I will focus, for most of the items considered in the empirical analysis, on estimates that rely on data stacked across several survey items rather than reporting estimates for single items. In the case where the estimation is based on stacked data, I thus focus on parameter estimates derived from a regression model with a slightly different structure:

$$y_{ict} = \alpha + \beta G_{it}^{\text{actual}} + \gamma x_{it} + \psi_{j[i]t} + \epsilon_{ict}, \qquad (2)$$

with  $y_{ict}$  denoting outcome y of individual i in year t and for item c belonging to some larger group of related items  $c = 1, \ldots, C$ ; with the total number of items C being larger than one (if there is only a single item, we are back to equation (1)). A first thing to note is that this procedure would – if all the items grouped together were available for exactly the same set of observations – scale up the original sample size by a factor exactly equal to the number of items grouped together.<sup>14</sup> However, in practice, because most of the items considered in the analysis do not appear in all four years of the survey (less than a quarter of all the items considered), because items appear in different rounds of the survey, and because of different nonresponse patterns across items, items which appear in more rounds of the survey will implicitly be given more weight than those that appear in fewer rounds (a similar argument applies to differences in the number of observations due to potentially different nonresponse patterns). The fact that relatively few items are observed in all four rounds of the survey is also the main reason why I cannot simply use average outcomes as the dependent variable – which would otherwise be the more natural way of aggregating several items together.

In those specifications where several items are stacked together in the way described above, I report standard errors clustered by individual to take the fact that the data contain repeated observations from the same person into account. At the same time, clustered standard errors will also correct for the inflation of the sample size due to stacking various outcomes from the same individuals (note that observations for different outcomes but the same individual have exactly the same values on all the regressors).<sup>15</sup>

<sup>&</sup>lt;sup>14</sup>For example, if there were three (i.e. C = 3) survey questions relating to the same conceptual subject, and if all three items were observed for exactly the same set of observations, then stacking the data would result in a sample exactly three times as large as the original sample size.

<sup>&</sup>lt;sup>15</sup>In fact, if the items were perfectly correlated with each other within any given individual, then this procedure would simply duplicate the information already available in the original data. This would mechanically deflate

The main practical advantage of this procedure is that it allows me to reduce the dimensionality of the data at hand considerably. Specifically, instead of reporting estimates for each of the 49 distinct items listed in table 1, the results reported below focus on 14 different outcomes only (these outcomes mostly represent groups of items or, in a few exceptional cases, single items).

## 4.3 Two-Stage Least Squares Estimates

One additional concern with simple OLS estimates of  $\beta$  based on equation (1) is the potential endogeneity of inequality perceptions due to attitudes about social inequality and inequality perceptions being determined simultaneously. Thus, in estimating the effect of inequality perceptions on attitudes, ideally one has to net out the potential reverse effect, running from attitudes to inequality perceptions.

To account for this potentially relevant issue, I will also present two-stage least squares (2SLS) estimates of the effect of inequality perceptions on individuals' attitudes to social inequality. I will use region-within-country means of subjective inequality perceptions as instrument for an individual's inequality perceptions.<sup>16</sup> The main argument underlying this instrument is that individual inequality perceptions are most likely influenced by the perceptions of people around them, such as their colleagues at work, their neighbors, or their relatives. At the same time, individual-level perceptions arguably have a negligible effect on mean perceptions in a given region. Even though somewhat less obvious, it might also be reasonable to assume that mean inequality perceptions have no direct effect on individual attitudes to social inequality.<sup>17</sup> The associated first-stage regression thus takes the following basic form:

$$G_{it}^{\text{actual}} = \pi_0 + \pi_1 \overline{G}_{k[i]t}^{\text{actual}} + \pi_2 x_{it} + \psi_{j[i]t} + \varepsilon_i, \qquad (3)$$

conventional standard errors, but standard errors clustered by individual would take properly account of this issue.

<sup>&</sup>lt;sup>16</sup>Dustmann and Preston (2001), for example, propose a similar logic for using an instrument at a higher level of spatial aggregation in their study on the effect of immigrants inflows on natives' attitudes to immigration.

<sup>&</sup>lt;sup>17</sup>That is, it is implicitly assumed that the instrument has only an indirect effect on the endogenous variable. In the context here, this assumption implies that region-within-country means of inequality perceptions have no direct effect on the outcome(s) other than through their effect on individual-level inequality perceptions. One potential concern with the specific instrument proposed here is that variation in the instrument might also reflect regional differences in the effective level of inequality (which are otherwise not taken into account in the analysis). In that case, the 2SLS estimates are likely to overestimate the effect of inequality perceptions on attitudes.

with  $G_{it}^{\text{actual}}$  denoting individual-level perceptions of inequality, and with  $\overline{G}_{k[i]t}^{\text{actual}}$  denoting mean inequality perceptions in region-within-country k and survey-year t, which is used as instrument for individual-level inequality perceptions. The remaining regressors in equation (3) are the same as those discussed in the context of equation (1) above. Specifically, note that it is still possible to control for country fixed-effects because there is within-country variation in the instrument (i.e. K > J). As usual,  $\pi_1$  and its associated standard error are informative about the strength of the instrument, and the tables below will report the F-statistic associated with the null hypothesis that the instrument has no partial effect on the endogenous variable.

# 5 Results

### 5.1 Beliefs about the Causes of Economic Success

The first set of estimates, shown in table 3, relates to individuals' beliefs about the causes of economic success, and I start with the results for individuals' beliefs about the reasons for getting ahead. In this and the following two tables, all estimates are based on groups of several survey items, as discussed in section 4.1 above (the exact number of items which are grouped together is noted in the bottom of each table).

### Table 3

The first five columns report estimates of the partial effect of inequality perceptions on people's belief that non-meritocratic mechanisms are important in determining individuals' wages. The following five columns report analogous estimates related to individuals' beliefs that meritocratic mechanisms determine differences in pay. In each of the two panels, the first three columns report OLS estimates, while the remaining two columns show the corresponding 2SLS estimates. For each point estimate reported (in this and all of the following tables), I also present (in brackets) an estimate of the approximate elasticity, evaluated at mean values, of the respective outcome measure with respect to individual-level inequality perceptions.

The estimate reported in the first column of table 3 is from a regression which includes the full set of country×survey-year fixed effects but no other controls. This first specification yields a positive and highly significant point estimate, implying that a higher perceived level of inequality is associated with a stronger belief that non-meritocratic mechanisms are important in determining wages in one's country of residence. The second column adds the two individuallevel controls (i.e. age and gender), resulting in a somewhat smaller, yet still significant estimate of inequality perceptions. However, both estimates imply a small effect, if judged by the implied elasticity. I add individual distributional norms as an additional individual-level regressor in the third column. This variable clearly appears to be important, as the point estimate on inequality perceptions from the third column is as much as 80% larger than the corresponding estimate from the specification that does not include distributional norms as a control variable (i.e. the point estimate increases from 0.08 to 0.146). The larger point estimate notwithstanding, however, note that the approximate elasticity remains quite small (equal to about 0.031).

Next, column 4 reports 2SLS estimates of the full specification, instrumenting inequality perceptions with region-within-country means of inequality perceptions.<sup>18</sup> This yields an even larger point estimate of about 0.188. However, this larger point estimate goes hand in hand with a much larger robust standard error, resulting in a statistically insignificant point estimate. Finally, I report 2SLS estimates in the fifth column in which I instrument for both inequality perceptions and distributional norms, with the instrument for the latter being the region-within-country mean of individual-level distributional norms (essentially mirroring the logic used to justify the instrument for individual inequality perceptions). The point estimate based on this final specification equals 0.173, close to the estimate from column 4, and implying an approximate elasticity of about 0.037. As in the preceding column, this point estimate comes with a comparatively large robust standard error of 0.142 and, consequently, this final point estimate also turns out to be statistically insignificant.<sup>19</sup>

The remaining five columns of table 3 report analogous estimates for individual's beliefs about the importance of meritocratic mechanisms in determining individual wages. There is a somewhat more interesting pattern of results here, insofar as the first two columns indicate that a higher level of inequality is associated with a stronger belief into meritocratic mecha-

<sup>&</sup>lt;sup>18</sup>The table also reports, in curly brackets, the F-statistic associated with the first-stage effect of the instrument(s) on the endogenous variable (i.e. individual-level inequality perceptions).

<sup>&</sup>lt;sup>19</sup>I have re-estimated the specifications reported in columns 1 to 5 of table 3, removing the two items with the lowest mean values. The two excluded items are the following: "A person's religion" and "the part of the country a person comes from". Estimates using the reduced set of items are shown in appendix table A.2. These additional estimates turn out to be larger than those using the full set of items, but with similar-sized standard errors (although they are based on a smaller sample).

nisms being important in determining individual pay, while the remaining three columns (which include distributional norms as a control) indicate that higher inequality perceptions induce a weaker belief into the importance of meritocratic mechanisms. Thus, the more demanding specifications point to a negative, though rather weak effect (implied elasticity of about -0.039 to -0.036) of inequality perceptions on the belief that meritocratic principles are important in determining pay.

#### Table 4

Moving on to individuals' beliefs about a potential functional role played by inequality, by setting incentives for economic agents, table 4 reports estimates quantifying how inequality perceptions impact individuals' beliefs about a potential incentivizing effect of inequality, both at the individual and the aggregate level.<sup>20</sup> The first three columns of table 4 show that there appears to be an effect of inequality perceptions on the belief that inequality serves as in incentive for individual effort. The two OLS estimates are small, and only the estimate from the specification that does not include distributional norms is statistically significant, but the 2SLS estimate turns out again to be significant and much larger than the corresponding OLS estimate. Also, this estimate implies a rather substantive elasticity of about 0.11. Interestingly, but also somewhat contradictory, the remaining three columns of table 4 show that there is a negative effect of inequality perceptions on the belief that inequality is necessary for an economy's success as a whole at the same time. Both OLS and 2SLS estimates are statistically significant, consistently negative, and also quite large in economic terms. Controlling for distributional norms, the point estimates imply an elasticity of -0.168 to -0.147.

#### Table 5

Additional complementary evidence on individuals' attitudes towards different pay norms is reported in table 5. As for individuals' beliefs about the reasons for getting ahead, it is possible to distinguish between meritocratic and non-meritocratic pay norms. Not surprisingly, the corresponding estimates indicate that higher inequality perceptions are associated with a

 $<sup>^{20}</sup>$ To save space, table 4 and all of the following tables only show estimates based on the key specifications (i.e. OLS estimates both with and without controlling for individual distributional norms as well as 2SLS estimates based on instrumenting both inequality perceptions and distributional norms); there are thus fewer columns from table 4 onwards than in table 3.

stronger belief in non-meritocratic pay norms, at least when distributional norms are held constant (note that the relevant OLS estimate switches its sign in this case, depending on whether distributional norms are held constant or not). Conditional on the inclusion of distributional norms, however, OLS and 2SLS estimates again appear consistent with each other (and again, 2SLS estimates turn out considerably larger than their OLS counterparts). Including distributional norms as a control variable also makes a difference regarding the effect of inequality perceptions on meritocratic pay norms. While the OLS estimate for the simpler specification turns out positive, it becomes negative once distributional norms are also taken into account. Moreover, note that OLS and 2SLS estimates appear inconsistent in this case, with the OLS estimate suggesting a negative effect, and the corresponding 2SLS estimate pointing to a positive effect of inequality perceptions on the importance of meritocratic pay norms.

# 5.2 The Perception of Chances and Risks Associated with Inequality

I next discuss estimates relating to the chances and risks that might be perceived as associated with a given level of inequality. Specifically, the first three columns of table 6 report estimates for individuals' beliefs that they stand a good chance of improving their standards of living. This item is clearly related to the outcomes considered in the preceding subsection, as it is very likely that individuals perceive their own chance of upward mobility to be less pervasive when they think that wages are substantively driven by non-meritocratic mechanisms. Arguably, it is also related to individuals' wage satisfaction, which is considered in the second panel of table 6 below.

Indeed, the estimates from the first three columns show that inequality perceptions are negatively and significantly associated with the belief that one stands a good chance of improving his or her standard of living. Also, in this case, the 2SLS estimate turns out to be of similar size as the corresponding OLS estimate using the full set of controls, and the estimates imply a relatively large sized effect in substantive terms (approximate elasticity of about -0.14 to -0.13, when controlling for distributional norms).

### Table 6

Moreover, given that individuals with high levels of perceived wage inequality appear to put considerably more emphasis on non-meritocratic determinants of wages, as shown in section 5.1 above, one would presumably also expect that these individuals are less satisfied with their own compensation. Not surprisingly, then, the remaining three columns of table 6 show that individual wage satisfaction is decreasing in the perceived level of wage inequality. As before, the OLS estimate is considerably larger when using the full set of controls, implying quite a large elasticity of wage satisfaction with respect to inequality perceptions of about -0.113. Instrumenting does not change the qualitative result, but it yields a much larger point estimate (the 2SLS estimate equals -0.875, compared to the corresponding OLS estimate of -0.419); which also implies a larger elasticity of about -0.237.

### Table 7

The first panel of table 7 shows estimates for people's overall normative assessment of inequality in their country of residence. Considering the results discussed so far, it is probably not surprising to find that inequality perceptions have a strong and highly significant effect on the individual perception that income differences in one's country are too large. In fact, both OLS and 2SLS estimates are statistically significant and they imply a comparatively large approximate elasticity of about 0.134 and 0.215, respectively, when holding distributional norms constant (note that the OLS estimate without controlling for distributional norms is still significant, but only about half the size of the OLS estimate including distributional norms as a control).

The second part of table 7 shows that individuals with higher inequality perceptions also tend to perceive a higher level of conflict in their society. Again, the inclusion of distributional norms turns out to be important for the qualitative result, as the OLS estimate without the inclusion yields an insignificant point estimate, but adding distributional norms turns the OLS estimate significant (point estimate of -0.003 and 0.176, respectively).

# 5.3 The Role of Government and Support of Progressive Taxation

The next set of estimates, presented in table 8, documents the effect of inequality perceptions on individuals' views about the appropriate role of the state with respect to distributional issues as well as to their support of progressive taxation.<sup>21</sup>

 $<sup>^{21}</sup>$ There is a closely related, voluminous empirical literature on the determinants of preferences for redistribution. See Alesina and Giuliano (2011) for a survey of this literature.

#### Table 8

Consistent with the previous results, the first three columns of table 8 show that a higher level of perceived inequality is statistically significantly associated with stronger support for government intervention. That is, people who perceive a high level of wage inequality appear to be considerably more supportive of intervention by the state. Both OLS and 2SLS estimates are statistically significant and positive and, again, including distributional norms as a control yields a much larger OLS estimate and instrumenting yields a still larger point estimate. When controlling for distributional norms, both OLS and 2SLS estimate imply a relatively large quantitative effect of inequality perceptions (as indicated by the comparatively large approximate elasticity of about 0.097 and 0.188, respectively).

The remaining three columns of table 8 show estimates on individuals' support of progressive taxation. Again, I find that there is a positive, and statistically highly significant, effect of inequality perceptions on the support of progressive taxation. The pattern is similar to the preceding results concerning the role of the state; with the OLS estimate approximately doubling when distributional norms are included as a control, and with a yet higher 2SLS estimate. Also, the economic size of the estimates is quite large, with an approximate elasticity about 0.11 to about 0.165, depending on the specification.

### 5.4 Political Preferences and Political Participation

The final set of estimates, reported in table 9, focuses on the effect of inequality perceptions on individuals' political preferences as well as their general political participation. The first three columns of table 9 report estimates for respondents' identification with the political right (that is, higher values on the dependent variable signify a stronger identity with the political right), while the remaining three columns focus on individuals' more general degree of political participation, measured by a single item asking about participation at the last election held in a respondent's country of residence.

#### Table 9

Not surprisingly, and consistent with the results presented so far, the first three columns of table 9 show that higher inequality perceptions are associated with people being significantly less (more) likely to state that they identify with the political right (left). Again, controlling for individual distributional norms, both OLS and 2SLS estimate imply quite a large effect of inequality perceptions on political preferences (with an approximate elasticity of about -0.102 and -0.309, respectively).

Finally, the evidence with respect to voting – as a measure of an individual's general political participation – is somewhat mixed. The two OLS estimates suggest that political participation is increasing in an individual's inequality perception, while the 2SLS estimate turns out to be small and statistically insignificant. If anything, then, a high perceived level of wage inequality seems to increase political participation.

# 6 Conclusions

In this paper, I use a simple yet intuitive empirical measure of individual-level inequality perceptions to estimate how these subjective perceptions affect individuals' attitudes to and beliefs about social inequality. Using a broad variety of different outcomes reflecting the diverse dimensions of attitudes to social inequality, the empirical analysis shows that a higher level of perceived wage inequality is statistically significantly associated with a weaker belief into the proper functioning of (labor) markets. Specifically, I find that individuals who perceive wage inequality to be high tend to be less (more) likely to believe that wage differentials are driven by meritocratic (non-meritocratic) principles. Consistent with previous empirical evidence, these individuals are also less likely to be satisfied with their own compensation from work. Taken together, these results suggest that a high perceived level of wage inequality has the potential to undermine the legitimacy of market outcomes. The results are also consistent with the argument that there is a feedback process running from (the perception of) inequality to attitudes to social inequality.

I further show that this result is robust to the use of an instrumental-variable estimation strategy which uses regional means of inequality perceptions as an instrument for individuallevel inequality perceptions. In general, 2SLS estimates turn out to be larger (in absolute terms) yet mostly consistent with the corresponding OLS estimates. Moreover, the implied approximate elasticities of attitudes and beliefs with respect to inequality perceptions reveal that the quantitative size of the estimates effects appear to be plausible. While most of the estimates describing the effect of inequality perceptions on attitudes to social inequality are statistically significant, most of the corresponding elasticities appear to be small or of moderate size.

Finally, the results presented in this paper also add to the available empirical evidence showing that subjective perceptions of economic phenomena shape individuals' attitudes and beliefs. The results of this paper show that attitudes to and beliefs about economic inequality are, in part, influenced by individuals' perception of the prevailing level of wage inequality in their country of residence. This in turn implies that changes in inequality might indeed bring about corresponding changes in attitudes towards social inequality.

# References

- Alesina, A. and Angeletos, G. (2005). Fairness and Redistribution. American Economic Review, 95(3), 960–980.
- Alesina, A. and Giuliano, P. (2011). Preferences for Redistribution. In Handbook of Social Economics, volume 1A, chapter 4. Elsevier.
- Alesina, A., Cozzi, G., and Mantovan, N. (2012). The evolution of ideology, fairness and redistribution. *Economic Journal*, **122**(565), 1244–1261.
- Angrist, J. D. and Pischke, J.-S. (2008). Mostly harmless econometrics: An empiricist's companion. Princeton University Press.
- Atkinson, A. B. (2008). The changing distribution of earnings in OECD countries. Oxford University Press.
- Bénabou, R. and Tirole, J. (2006). Belief in a just world and redistributive politics. Quarterly Journal of Economics, 121(2), 699–746.
- Bjørnskov, C., Dreher, A., Fischer, J. A., Schnellenbach, J., and Gehring, K. (2013). Inequality and happiness: When perceived social mobility and economic reality do not match. *Journal* of Economic Behavior & Organization, **91**, 75–92.
- Card, D., Mas, A., Moretti, E., and Saez, E. (2012). Inequality at Work: The Effect of Peer Salaries on Job Satisfaction. *American Economic Review*, **102**(6), 2981–3003.
- Chambers, J. R., Swan, L. K., and Heesacker, M. (2014). Better off than we know distorted perceptions of incomes and income inequality in America. *Psychological Science*, 25(2), 613–618.
- Charness, G. and Kuhn, P. (2007). Does pay inequality affect worker effort? Experimental evidence. *Journal of Labor Economics*, **25**(4), 693–723.
- Clark, A. and Oswald, A. (1996). Satisfaction and comparison income. Journal of Public Economics, 61(3), 359–381.
- Clark, A. and Senik, C. (2010). Who compares to whom? The anatomy of income comparisons in Europe. *Economic Journal*, **120**(544), 573–594.
- Clark, A. E. and D'Ambrosio, C. (2015). Attitudes to Income Inequality: Experimental and Survey Evidence. In *Handbook of Income Distribution*, volume 2A, chapter 13, pages 1147– 1208. North-Holland.
- Clark, A. E., Westergård-Nielsen, N., and Kristensen, N. (2009). Economic satisfaction and income rank in small neighbourhoods. *Journal of the European Economic Association*, 7(2/3), 519–527.
- Clark, A. E., Masclet, D., and Villeval, M. C. (2010). Effort and comparison income: Experimental and survey evidence. *Industrial & Labor Relations Review*, **63**(3), 407–426.
- Cornelissen, T., Himmler, O., and Koenig, T. (2011). Perceived unfairness in CEO compensation and work morale. *Economics Letters*, **110**(1), 45–48.

- Cruces, G., Truglia, R., and Tetaz, M. (2013). Biased perceptions of income distribution and preferences for redistribution: Evidence from a survey experiment. *Journal of Public Economics*, 98, 100–112.
- Di Tella, R., Galiani, S., and Schargrodsky, E. (2007). The formation of beliefs: Evidence from the allocation of land titles to squatters. *Quarterly Journal of Economics*, **122**(1), 209–241.
- Dustmann, C. and Preston, I. (2001). Attitudes to ethnic minorities, ethnic context and location decisions. *Economic Journal*, **111**(470), 353–373.
- Ferrer-i-Carbonell, A. (2005). Income and well-being: an empirical analysis of the comparison income effect. Journal of Public Economics, 89(5-6), 997–1019.
- Gemmell, N., Morrissey, O., and Pinar, A. (2004). Tax perceptions and preferences over tax structure in the United Kingdom. *Economic Journal*, **114**(493), F117–F138.
- Giuliano, P. and Spilimbergo, A. (2014). Growing up in a recession. Review of Economic Studies, 81(2), 787–817.
- ISSP Research Group (2014a). International Social Survey Programme: Social Inequality I-IV - ISSP 1987-1992-1999-2009. GESIS Data Archive, Cologne. ZA5890 Data File Version 1.0.0.
- ISSP Research Group (2014b). International Social Survey Programme: Social Inequality I-IV add on ISSP 1987-1992-1999-2009. GESIS Data Archive, Cologne. ZA5891 Data file Version 1.1.0.
- Kuhn, A. (2011). In the eye of the beholder: Subjective inequality measures and individuals' assessment of market justice. *European Journal of Political Economy*, **27**(4), 625–641.
- Kuhn, A. (2015). The Individual Perception of Wage Inequality: A Measurement Framework and Some Empirical Evidence. IZA Discussion Paper No. 9579.
- Kuziemko, I., Norton, M. I., and Saez, E. (2015). How elastic are preferences for redistribution? evidence from randomized survey experiments. *American Economic Review*, **105**(4), 1478– 1508.
- Luttmer, E. (2005). Neighbors as Negatives: Relative Earnings and Well-Being. *Quarterly Journal of Economics*, **120**(3), 963–1002.
- Norton, M. and Ariely, D. (2011). Building a better America one wealth quintile at a time. *Perspectives on Psychological Science*, **6**(1), 9–12.
- Olken, B. (2009). Corruption perceptions vs. corruption reality. *Journal of Public Economics*, **93**(7), 950–964.
- Osberg, L. and Smeeding, T. (2006). "Fair" inequality? Attitudes toward pay differentials: the United States in comparative perspective. *American Sociological Review*, **71**(3), 450–473.
- Schneider, S. (2012). Income inequality and its consequences for life satisfaction: What role do social cognitions play? *Social Indicators Research*, **106**(3), 419–438.
- Schwarze, J. and Härpfer, M. (2007). Are people inequality averse, and do they prefer redistribution by the state? Evidence from German longitudinal data on life satisfaction. *Journal of Socio-Economics*, **36**(2), 233–249.

- Senik, C. (2005). Income distribution and well-being: what can we learn from subjective data? Journal of Economic Surveys, **19**(1), 43–63.
- Verme, P. (2011). Life satisfaction and income inequality. *Review of Income and Wealth*, **57**(1), 111–127.
- Winkelmann, L. and Winkelmann, R. (2010). Does Inequality Harm the Middle Class? *Kyklos*, **63**(2), 301–316.

	Mean	Sd	Min	Max	n	Τ
(a) Reasons for getting ahead						
(i) Non-meritocratic principles						
Coming from a wealthy family	2.847	1.136	1	5	85,010	4
Having well-educated parents	3.082	1.036	1	5	57,118	3
Knowing the right people	3.484	1.006	1	5	85,361	4
Having political connections	2.568	1.165	1	5	$56,\!597$	į
A person's race	2.221	1.119	1	5	$53,\!116$	:
A person's religion	1.868	1.029	1	5	55,763	:
The part of the country a person comes from	1.976	0.997	1	5	28,719	4
Being born a man or a woman	2.214	1.120	1	5	$55,\!855$	
A person's political beliefs	2.234	1.058	1	5	27,777	4
Have to be corrupt	2.739	1.273	1	5	$53,\!369$	4
(ii) Meritocratic principles						
Having a good education oneself	3.876	0.907	1	5	58,967	:
Having ambition	3.962	0.888	1	5	58,523	
Having natural ability	3.732	0.864	1	5	$30,\!477$	4
Hard work	3.934	0.915	1	5	$58,\!813$	;
(b) Incentivizing effect of inequality						
(i) At the individual level						
To get people to work hard	2.891	0.851	1	4	$11,\!675$	4
No extra responsibility at work	3.811	1.017	1	5	30,344	4
No objections to extra qualification	3.677	1.070	1	5	$30,\!292$	4
No one would study for years	3.794	1.106	1	5	56,774	
(ii) At the aggregate level						
Large differences necessary for prosperity	2.575	1.159	1	5	54,803	ę
Allowing business to make profits	3.256	1.129	1	5	$27,\!378$	4
(c) Pay norms ("what should be important for	· pay")					
(i) Non-meritocratic norms						
What is needed to support a family	3.438	1.073	1	5	72,321	
Whether someone has children to support	3.263	1.151	1	5	72,429	
(ii) Meritocratic norms						
Number of years spent in education and training	3.579	0.897	1	5	$73,\!275$	÷
Whether the job requires supervising others	3.477	0.897	1	5	44,811	, ,
How well someone does the job	4.156	0.732	1	$\frac{5}{5}$	72,711	
How hard someone works	4.023	0.792	1	5	73,265	

Table 1: Descriptive statistics, outcome measures

	Mean	$\operatorname{Sd}$	Min	Max	n	Т
(d) Chances and risks associated with inequality						
(i) Improving one's living standard						
Good chance of improving standard of living	2.987	1.124	1	5	27,779	2
(ii) Wage satisfaction						
Earn less or more than deserved How just is your pay compared to own skills and efforts	$2.228 \\ 2.191$	$0.796 \\ 0.834$	1 1	$5\\5$	49,467 50,622	$2 \\ 2$
(iii) Normative assessment of inequality						
Income differences are too large	4.126	0.969	1	5	85,125	4
(iv) Perception of conflicts in society						
Poor people - rich people Working class - middle class	$2.492 \\ 2.026$	$0.855 \\ 0.748$	1 1	$4 \\ 4$	$82,510 \\ 81,090$	$\frac{4}{4}$
Unemployed people - employed people	2.314	0.844	1	4	$25,\!184$	2
Management - workers Farmers - city people People at the top of the society - people at the bottom	$2.546 \\ 2.057 \\ 2.679$	$0.800 \\ 0.827 \\ 0.936$	1 1 1	4 $4$ $4$	81,510 29,048 52,560	$     \begin{array}{c}       4 \\       2 \\       2     \end{array} $
Young people - older people	2.204	0.804	1	4	54,243	$\frac{2}{3}$
(e) The role of government and support of progr	ressive ta	axation				
(i) The role of government						
Reduce income differences Provide chances for poor children to go to university Provide jobs for everyone who wants one Provide decent living standard for unemployed Spend less on benefits for the poor (-)	$\begin{array}{c} 3.710 \\ 4.023 \\ 3.846 \\ 3.697 \\ 2.238 \end{array}$	$1.172 \\ 0.925 \\ 1.156 \\ 1.045 \\ 1.103$	1 1 1 1	5 5 5 5 5	84,151 12,104 30,367 36,110 38,217	$4 \\ 2 \\ 3 \\ 3$
Provide basic income for all	3.635	1.233	1	5	28,613	2
(ii) Support of progressive taxation						
People with high incomes should pay more taxes Taxes for those with high incomes	$4.019 \\ 3.337$	$0.773 \\ 1.090$	1 1	$5\\5$	$\begin{array}{c} 80,\!642 \\ 53,\!107 \end{array}$	$\frac{4}{3}$
(f) Political preferences and political participation	n					
(i) Political identification (with the political right)						
Party affiliation (affiliation to a certain party) Party affiliation (question on left-right placement) Party voted in last election	2.907 2.813 3.007	$0.966 \\ 1.029 \\ 1.004$	$egin{array}{c} 1 \\ 1 \\ 1 \end{array}$	5 5 5	$33,442 \\ 12,555 \\ 14,720$	$4\\3\\4$
(ii) Political participation						
Voted in last election (yes $= 1$ )	0.794	0.405	0	1	47,516	4

Table 1: Continued

Notes: The table shows descriptives for all the items eventually used as an outcome. The first four columns show, respectively, the mean, the standard deviation, as well as the sample minimum and maximum (note that all variables only take on integer values). The last two columns show the number of observations with valid values (n) as well as the number of years (out of a maximum of four) a given item appeared in the survey (T).

		TTOTOTAOD	COI LEIAUIOII
(a) Individual-level inequality perceptions			
Inequality perception	0.560	0.218	0.395
<b>1</b> (inequality perception < 0)	0.004	0.067	
<b>1</b> (inequality perception = 0)	0.001	0.034	
<b>1</b> (inequality perception > 0)	0.994	0.075	
(b) Other individual-level controls			
Age (in years)	44.957	16.361	0.037
Female (yes $= 1$ )	0.519	0.500	0.007
Distributional norms	0.388	0.209	0.287

Table 2: Descriptive statistics, inequality perceptions and other individual-level controls

g ahead
getting
$\operatorname{for}$
Reasons
3:
Table

		Non-me	Non-meritocratic mechanisms	nechanisms			Mer	Meritocratic mechanisms	chanisms	
Mean Standard deviation	2.636 1.213	2.636 1.213	2.636 1.213	$2.636 \\ 1.213$	2.636 1.213	$3.896 \\ 0.901$	$3.896 \\ 0.901$	$3.896 \\ 0.901$	$3.896 \\ 0.901$	$3.896 \\ 0.901$
Inequality perception $G_i^{\text{actual}}$	$\begin{array}{c} 0.107^{***} \\ (0.015) \\ [0.023] \end{array}$	0.080*** (0.015) [0.017]	$\begin{array}{c} 0.146^{\star\star\star} \\ (0.018) \\ [0.031] \end{array}$	$\begin{array}{c} 0.188\\ (0.142)\\ [0.040]\\ \{899.923\} \end{array}$	$\begin{array}{c} 0.173\\ (0.120)\\ [0.037]\\ \{748.460\} \end{array}$	$\begin{array}{c} 0.091^{***} \\ (0.016) \\ [0.013] \end{array}$	$0.083^{***}$ (0.016) [0.012]	$-0.053^{***}$ (0.019) [-0.007]	$\begin{array}{c} -0.278^{*} \\ (0.162) \\ [-0.039] \\ \{668.692\} \end{array}$	$\begin{array}{c} -0.254^{\star} \\ (0.138) \\ [-0.036] \\ \{569.080\} \end{array}$
Estimation method Country×year FEs	OLS Yes	OLS Yes	OLS Yes	2SLS Yes	2SLS Yes	OLS Yes	OLS Yes	OLS Yes	2SLS Yes	2SLS Yes
Individual-level controls Distributional norms	No No	${ m Yes}_{ m No}$	${ m Yes}{ m Yes}$	${ m Yes}{ m Yes}$	$\substack{\mathrm{Yes}}{\mathrm{Yes}}$	No No	${ m Yes}_{ m No}$	${ m Yes}$	${ m Yes}{ m Yes}$	${ m Yes}{ m Yes}$
Number of observations p-value (F-statistic) R-Squared	539,035 0.000 0.072	539,035 0.000 0.073	539,035 0.000 0.073	539,035 0.000 0.073	539,035 0.000 0.073	$\begin{array}{c} 199,516 \\ 0.000 \\ 0.047 \end{array}$	$\begin{array}{c} 199,516\\ 0.000\\ 0.047\end{array}$	$199,516 \\ 0.000 \\ 0.048$	$\begin{array}{c} 199,516\\ 0.000\\ 0.047\end{array}$	$199,516 \\ 0.000 \\ 0.047$
Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated with the null hypothesis of no partial effect of the instrument(s) on individual-level inequality perceptions is reported in curly brackets. The estimates reported in columns 1 to 5 (6 to 10) use data stacked across 10 (4) different items (see table 1 for details).	cal significan ependent var bartial effect a stacked aci	ce on the 10 <sup>6</sup> iable with re of the instru ross 10 (4) di	%, 5%, and spect to ineq ment(s) on i ifferent items	1% level, respe uality percept ndividual-leve ; (see table 1 f	nd 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated on individual-level inequality perceptions is reported in curly brackets. The estimates reported in tems (see table 1 for details).	ard errors ar in brackets. I reptions is	e clustered l n the case o reported in	by individual i f 2SLS estima curly brackets	and are given i tes, the F-stati s. The estimat	n parenthese stic associate es reported

	At t]	At the individual level	al level	At t	At the aggregate level	level
Mean Standard deviation	$3.691 \\ 1.087$	$\begin{array}{c} 3.691 \\ 1.087 \end{array}$	$3.691 \\ 1.087$	$2.801 \\ 1.194$	$2.801 \\ 1.194$	$2.801 \\ 1.194$
Inequality perception $G_i^{\text{actual}}$	$0.049^{**}$ (0.023) [0.007]	$\begin{array}{c} 0.033 \\ (0.028) \\ [0.005] \end{array}$	$\begin{array}{c} 0.804^{\star\star\star}\\ (0.231)\\ [0.110]\\ [347.396]\end{array}$	$-0.340^{***}$ (0.029) [-0.064]	$-0.891^{***}$ (0.035) [-0.168]	$\begin{array}{c} -0.777 *** \\ (0.259) \\ [-0.147] \\ \{377.513\} \end{array}$
Estimation method Country×vear FEs	OLS Yes	OLS Yes	2SLS Yes	OLS Yes	OLS Yes	2SLS Yes
Individual-level controls Distributional norms	${ m Yes} { m No}$	$\mathop{\rm Yes}\limits_{\mathop{\rm Yes}}$	$\substack{\text{Yes}}{\text{Yes}}$	$\mathop{\rm Yes}\limits_{\rm No}$	$\mathop{\rm Yes}\limits_{\mathop{\rm Yes}}$	$\substack{\text{Yes}}{\text{Yes}}$
Number of observations p-value (F-statistic) R-Squared	$\frac{124,357}{0.000}$ 0.058	$\begin{array}{c} 124,357\\ 0.000\\ 0.058\end{array}$	$\begin{array}{c} 124,357 \\ 0.000 \\ 0.047 \end{array}$	$79,216 \\ 0.000 \\ 0.088$	79,216 0.000 0.102	$\begin{array}{c} 79,216\\ 0.000\\ 0.101\end{array}$
Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated with the null hypothesis of no partial effect of the instrument(s) on individual-level inequality perceptions is reported in curly brackets. The estimates reported in columns 1 to 3 (4 to 6) use data stacked across 4 (2) different items (see table 1 for details).	tical significa are given in F are given in b effect of the reported in	unce on the parentheses. prackets. In t instrument( columns 1 to	10%, $5%$ , and Approximate he case of $2SL$ s) on individu 3 (4 to 6) us	<ul> <li>1 % level, re elasticities of S estimates, th ial-level inequi e data stackee</li> </ul>	spectively. St the dependent he F-statistic a ality perceptio a across 4 (2) (	andard errors variable with ssociated with ns is reported different items

Table 4: Inequality as an incentive device

	Non-m	Non-meritocratic norms	norms	Μ	Meritocratic norms	Ims
Mean Standard deviation	$3.350 \\ 1.116$	$3.350 \\ 1.116$	$3.350 \\ 1.116$	$3.845 \\ 0.873$	$3.845 \\ 0.873$	$3.845 \\ 0.873$
Inequality perception $G_i^{\text{actual}}$	$-0.262^{***}$ (0.022) [-0.045]	$\begin{array}{c} 0.170^{\star\star\star} \\ (0.026) \\ [0.029] \end{array}$	$\begin{array}{c} 0.471^{\star\star\star}\\ (0.177)\\ [0.082]\\ \{701.142\}\end{array}$	$0.028^{\star\star}$ (0.013) [0.004]	$-0.072^{***}$ (0.015) [-0.011]	$\begin{array}{c} 0.198^{\star} \\ (0.104) \\ [0.030] \\ \{670.901\} \end{array}$
Estimation method Country×year FEs Individual-level controls Distributional norms	OLS Yes No	OLS Yes Yes	2SLS Yes Yes	OLS Yes No	OLS Yes Yes	2SLS Yes Yes Yes
Number of observations p-value (F-statistic) R-Squared	$139,721 \\ 0.000 \\ 0.097$	$139,721 \\ 0.000 \\ 0.109$	139,721 0.000 0.098	254,736 0.000 0.036	$254,736 \\ 0.000 \\ 0.037$	254,736 0.000 0.035
Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated with the null hypothesis of no partial effect of the instrument(s) on individual-level inequality perceptions is reported in curly brackets. The estimates reported in columns 1 to 3 (4 to 6) use data stacked across 2 (4) different items (see table 1 for details).	ical significan re given in pa are given in bre effect of the i reported in cc	ce on the 10 rentheses. A ackets. In th astrument(s) blumns 1 to 5	0%, $5%$ , and pproximate el e case of 2SLS ) on individua 3 (4 to 6) use	1% level, re asticities of estimates, th l-level inequi data stacked	spectively. St the dependent as F-statistic a ality perceptio I across 2 (4) c	andard errors variable with ssociated with is is reported ifferent items

Table 5: Pay norms

	Improvi	Improving standard of living	of living	Μ	Wage satisfaction	on
Mean Standard deviation	$2.985 \\ 1.126$	2.985 1.126	2.985 1.126	$2.207 \\ 0.816$	$2.207 \\ 0.816$	$2.207 \\ 0.816$
Inequality perception $G_i^{\text{actual}}$	-0.249*** (0.040) [-0.043]	$-0.814^{***}$ (0.050) [-0.142]	$\begin{array}{c} -0.753 \\ (0.442) \\ [-0.131] \\ \{178.153\} \end{array}$	$-0.155^{***}$ (0.020) [-0.042]	$-0.419^{***}$ (0.023) [-0.113]	$\begin{array}{c} -0.875^{\star\star\star} \\ (0.143) \\ [-0.237] \\ [569.514] \end{array}$
Estimation method Country×year FEs Individual-level controls Distributional norms	OLS Yes No	$\begin{array}{c} \mathrm{OLS} \\ \mathrm{Yes} \\ \mathrm{Yes} \\ \mathrm{Yes} \end{array}$	2SLS Yes Yes	OLS Yes No	OLS Yes Yes Yes	2SLS Yes Yes
Number of observations p-value (F-statistic) R-Squared	$\begin{array}{c} 26,777\\ 0.000\\ 0.165\end{array}$	26,777 26,777 0.000 0.181	26,777 26,777 0.000 0.172	96,590 0.000 0.078	96,590 0.000 0.087	96,590 0.000 0.067
Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated with the null hypothesis of no partial effect of the instrument(s) on individual-level inequality perceptions is reported in curly brackets. The estimates reported in columns 1 to 3 (4 to 6) use data stacked across 1 (2) different items (see table 1 for details).	ical significanc iven in parentl en in brackets et of the instru in columns 1	te on the $10\%$ aeses. Approx . In the case ment(s) on in to 3 (4 to 6) t	6, 5%, and 1% imate elasticit of 2SLS estir dividual-level use data stack	5 level, respecties of the deptiles of the deptiles, the F-s mates, the F-s mequality per defactors 1 (2)	tively. Stand endent variable statistic associ ceptions is rep ) different iten	ard errors are e with respect ated with the orted in curly is (see table 1

Table 6: Standing a chance of improving one's standard of living and wage satisfaction

	Ineq	Inequality is too large	large	Cor	Conflicts in society	iety
Mean Standard deviation	$4.132 \\ 0.968$	$4.132 \\ 0.968$	$4.132 \\ 0.968$	2.353 0.859	2.353 $0.859$	2.353 0.859
Inequality perception $G_i^{\text{actual}}$	$\begin{array}{c} 0.382^{\star\star\star} \\ (0.019) \\ [0.052] \end{array}$	$\begin{array}{c} 0.988^{\star\star\star} \\ (0.023) \\ [0.134] \end{array}$	$\begin{array}{c} 1.583^{\star\star\star\star}\\ (0.142)\\ [0.215]\\ \{790.215\}\end{array}$	$\begin{array}{c} -0.003 \\ (0.013) \\ [-0.001] \end{array}$	$\begin{array}{c} 0.176^{\star\star\star} \\ (0.015) \\ [0.042] \end{array}$	0.483*** (0.108) [0.114] {751.236}
Estimation method Country×vear FEs	OLS Yes	OLS Yes	2SLS Yes	OLS Yes	OLS Yes	2SLS Yes
Individual-level controls Distributional norms	$\mathop{\rm Yes}_{\rm No}$	$\substack{\mathrm{Yes}}{\mathrm{Yes}}$	$\mathop{\rm Yes}\limits_{\mathop{\rm Yes}}$	$\mathop{\rm Yes}_{\rm No}$	$\mathop{\rm Yes}\limits_{\mathop{\rm Yes}}$	$\substack{\text{Yes}}{\text{Yes}}$
Number of observations p-value (F-statistic) R-Squared	82,116 0.000 0.153	82,116 0.000 0.183	$\begin{array}{c} 82,116\\ 0.000\\ 0.175\end{array}$	392,322 0.000 0.076	392,322 0.000 0.079	392,322 0.000 0.074
Notes: $*$ , $**$ , denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated with the null hypothesis of no partial effect of the instrument(s) on individual-level inequality perceptions is proved in columns 1 to 3 (4 to 6) used data structed arrors of 1 (7)	al significanc given in par are given in artial effect	the investment of the investment of the instru- of the instru-	6, 5%, and 1% proximate els the case of 2 ment(s) on i	6 level, respectivities of the stricties of the SLS estimates individual-leventee of the strict of t	tively. Stand e dependent , the F-statis	ard errors arv variable with tic associated perceptions i

different items (see table 1 for details).

Table 7: The normative assessment of inconality and the perception of conflicts in society

	The 1	The role of government	ment	Support	Support of progressive taxation	e taxation
Mean Standard deviation	$\begin{array}{c} 3.493 \\ 1.269 \end{array}$	3.493 $1.269$	3.493 $1.269$	$3.749 \\ 0.970$	$\begin{array}{c} 3.749\\ 0.970\end{array}$	$3.749 \\ 0.970$
Inequality perception $G_i^{\text{actual}}$	$0.034^{*}$ (0.018) [0.005]	$\begin{array}{c} 0.662^{\star\star\star} \\ (0.022) \\ [0.097] \end{array}$	1.281*** (0.148) [0.188] {718.611}	$\begin{array}{c} 0.334^{\star\star\star} \\ (0.017) \\ [0.051] \end{array}$	$\begin{array}{c} 0.726^{\star\star\star} \\ (0.020) \\ [0.110] \end{array}$	$\begin{array}{c} 1.088^{\star\star\star}\\ (0.135)\\ [0.165]\\ [0.165]\\ \{794.703\}\end{array}$
Estimation method Country×year FEs Individual-level controls Distributional norms	OLS Yes No	OLS Yes Yes Yes	2SLS Yes Yes Yes	OLS Yes No	OLS Yes Yes Yes	2SLS Yes Yes Yes
Number of observations p-value (F-statistic) R-Squared	221,515 0.000 0.104	221,515 0.000 0.124	$\begin{array}{c} 221,515\\ 0.000\\ 0.110\end{array}$	$\begin{array}{c} 129,278 \\ 0.000 \\ 0.096 \end{array}$	$\begin{array}{c} 129,278 \\ 0.000 \\ 0.108 \end{array}$	$\begin{array}{c} 129,278 \\ 0.000 \\ 0.100 \end{array}$
Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated with the null hypothesis of no partial effect of the instrument(s) on individual-level inequality perceptions is reported in curly brackets. The estimates reported in columns 1 to 3 (4 to 6) use data stacked across 6 (2) different items (see table 1 for details).	al significanc jiven in pare are given in artial effect e estimates re tails).	e on the 10% ntheses. App brackets. In 1 of the instru ported in col	, 5%, and 1% proximate elas the case of 2S ment(s) on in humns 1 to 3	level, respec ticities of th LS estimates dividual-leve (4 to 6) use	tively. Stand te dependent , the F-statis I inequality J data stacked	ard errors are variable with tic associated berceptions is l across 6 (2)

Table 8: The role of government and support of progressive taxation

	Poli	Political identification	ation	Polit	Political participation	pation
Mean Standard deviation	$2.910 \\ 0.991$	$2.910 \\ 0.991$	$2.910 \\ 0.991$	$0.794 \\ 0.404$	$0.794 \\ 0.404$	$0.794 \\ 0.404$
Inequality perception $G_i^{\text{actual}}$	-0.083*** (0.028) [-0.017]	$-0.507^{***}$ (0.034) [-0.102]	$\begin{array}{c} -1.537 *** \\ (0.244) \\ [-0.309] \\ [399.492] \end{array}$	$\begin{array}{c} 0.122^{\star\star\star} \\ (0.012) \\ [0.091] \end{array}$	$\begin{array}{c} 0.052^{\star\star\star} \\ (0.014) \\ [0.038] \end{array}$	$\begin{array}{c} 0.029\\ (0.089)\\ [0.022]\\ \{576.093\}\end{array}$
Estimation method Country×year FEs	OLS Yes	OLS Yes	2SLS Yes	OLS Yes	OLS Yes	2SLS Yes
Individual-level controls Distributional norms	Yes No	Yes Yes	${ m Yes}$	${ m Yes}_{ m No}$	${ m Yes}{ m Yes}$	${ m Yes}{ m Yes}$
Number of observations p-value (F-statistic) R-Squared	58,587 0.000 0.044	58,587 0.000 0.056	58,587 0.000 0.034	$\begin{array}{c} 45,886\\ 0.000\\ 0.150\end{array}$	$\begin{array}{c} 45,886\\ 0.000\\ 0.152\end{array}$	$\begin{array}{c} 45,886\\ 0.000\\ 0.152\end{array}$
Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated with the null hypothesis of no partial effect of the instrument(s) on individual-level inequality perceptions is reported in curly brackets. The estimates reported in columns 1 to 3 (4 to 6) use data stacked across 3 (1) different items (see table 1 for details).	tical significant tre given in para re given in br effect of the i reported in c	(ce on the 10%) trentheses. Ap ackets. In the a nstrument(s) of olumns 1 to 3	%, 5%, and 1% proximate elas case of 2SLS es on individual-le (4 to 6) use da	6 level, resp ticities of th timates, the evel inequali ta stacked a	ectively. Sta e dependent F-statistic as ty perception cross 3 (1) d	andard errors variable with ssociated with as is reported lifferent items

Table 9: Political preferences and political participation

# A Additional tables

			Numbe	r of obser	rvations	
		1987	1992	1999	2009	Total
Australia	AU	1,442	1,997	1,580	$1,\!375$	6,39
Austria	AT	902		895	988	2,78
Bulgaria	BG		949	821	556	2,320
Canada	CA		913	932		$1,\!84!$
Chile	$\operatorname{CL}$			1,078	$1,\!407$	$2,\!48!$
Cyprus	CY			$1,\!000$	974	1,974
Czech Republic	CZ		656	$1,\!698$	1,160	$^{3,51}$
France	$\mathbf{FR}$			1,748	$2,\!371$	$4,\!119$
Germany	DE	$1,\!282$	$2,\!992$	$1,\!223$	1,208	6,703
Hungary	HU	2,269	$1,\!103$	1,014	870	$5,\!250$
Israel	IL			$1,\!184$	$1,\!071$	2,258
Italy	IT		979		992	1,97
Japan	$_{\rm JP}$			685	697	1,382
Latvia	LV			1,064	919	1,983
New Zealand	NZ		$1,\!151$	1,019	875	$3,\!04!$
Norway	NO		$1,\!424$	$1,\!193$	$1,\!404$	4,02
Philippines	$\mathbf{PH}$		911	$1,\!085$	1,021	3,01'
Poland	PL	$1,\!560$	$1,\!357$	910	988	4,813
Portugal	$\mathbf{PT}$			1,014	710	1,72
Russia	RU		$1,\!698$	$1,\!277$	$1,\!436$	4,41
Slovak Republic	SK		400	$1,\!061$	$1,\!100$	2,561
Slovenia	$\mathbf{SI}$		964	942	876	2,782
Spain	$\mathbf{ES}$			970	978	1,948
Sweden	SE		678	1,043	$1,\!051$	2,772
Switzerland	$\operatorname{CH}$	879			$1,\!125$	2,004
United Kingdom	UK	1,046	959	666	853	$3,\!52$
United States	US	$1,\!317$	$1,\!182$	1,029	$1,\!450$	4,978
Total		10,697	20,313	27,131	28,455	86,59

Table A.1: Number of observations, by country and year

Notes: The table shows the number of observations with nonmissing inequality perceptions.

			Full set of items	ems			Res	Restricted set of items	of items	
Mean Standard deviation	2.636 1.213	2.636 $1.213$	2.636 1.213	2.636 1.213	2.636 1.213	$2.767 \\ 1.199$	$\begin{array}{c} 2.767 \\ 1.199 \end{array}$	$2.767 \\ 1.199$	2.767 1.199	2.767 1.199
Inequality perception $G_i^{\text{actual}}$	0.107*** (0.015) [0.023]	0.080*** (0.015) [0.017]	$\begin{array}{c} 0.146^{\star\star\star} \\ (0.018) \\ [0.031] \end{array}$	$\begin{array}{c} 0.188\\ (0.142)\\ [0.040]\\ 899.923 \end{array}$	$\begin{array}{c} 0.173\\ (0.120)\\ [0.037]\\ [748.460]\end{array}$	0.150*** (0.015) [0.031]	$\begin{array}{c} 0.128^{\star\star\star} \\ (0.015) \\ [0.026] \end{array}$	0.204*** (0.018) [0.042]	$\begin{array}{c} 0.280^{\star} \\ (0.145) \\ [0.057] \\ 942.021 \end{array}$	$\begin{array}{c} 0.276^{\star\star}\\ (0.123)\\ [0.057]\\ [0.057]\\ \{773.886\}\end{array}$
Estimation method Country×year FEs Individual-level controls Distributional norms	OLS Yes No No	$\begin{array}{c} \text{OLS} \\ \text{Yes} \\ \text{Yes} \\ \text{No} \end{array}$	OLS Yes Yes Yes	2SLS Yes Yes Yes	2SLS Yes Yes Yes	OLS Yes No No	$\begin{array}{c} \text{OLS} \\ \text{Yes} \\ \text{Yes} \\ \text{No} \end{array}$	OLS Yes Yes Yes	$^{2\mathrm{SLS}}_{\mathrm{Yes}}$ Yes Yes	2SLS Yes Yes Yes
Number of observations p-value (F-statistic) R-Squared	539,035 0.000 0.072	539,035 0.000 0.073	539,035 0.000 0.073	539,035 0.000 0.073	539,035 0.000 0.073	$\begin{array}{c} 457,484 \\ 0.000 \\ 0.063 \end{array}$	$\begin{array}{c} 457,484 \\ 0.000 \\ 0.063 \end{array}$	$\begin{array}{c} 457,484 \\ 0.000 \\ 0.064 \end{array}$	$\begin{array}{c} 457,484 \\ 0.000 \\ 0.064 \end{array}$	457,484 0.000 0.064
Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. In the case of 2SLS estimates, the F-statistic associated with the null hypothesis of no partial effect of the instrument(s) on individual-level inequality perceptions is reported in curly brackets. The estimates reported in columns 1 to 5 (6 to 10) use data stacked across 10 (8) different items (see table 1 for details).	al significanc pendent vari urtial effect o stacked acr	e on the 10% able with res f the instrur oss 10 (8) di	6, 5%, and 1 <sup>(</sup> pect to inequ nent(s) on ir fferent items	% level, respectability percepti dividual-level (see table 1 f	ctively. Standa ons are given ir inequality per or details).	rd errors are 1 brackets. In ceptions is r	clustered by 1 the case of eported in c	individual 2SLS estima 1rly bracket	and are given tes, the F-stat s. The estima	in parenthese istic associate tes reported i

Table A.2: Reasons for getting ahead (non-meritocratic mechanisms only), full versus restricted set of items

	Z	on-meritoc	ratic reason	ns important	Non-meritocratic reasons important for getting ahead	ead
Mean	2.636	2.628	2.628	2.636	2.628	2.628
Standard deviation	1.213	1.211	1.211	1.213	1.211	1.211
Inequality perception	$0.146^{***}$	0.150***	0.143***	0.173	0.191	0.115
	(0.018)	(0.020)	(0.020)	(0.120)	(0.132)	(0.134)
	[0.031]	[0.033]	[0.031]	[0.037]	[0.042]	[0.025]
				$\{748.460\}$	$\{609.237\}$	$\{608.710\}$
Estimation method	OLS	OLS	OLS	2SLS	2SLS	2SLS
Country×year FEs	$\mathbf{Yes}$	Yes	Yes	$\mathbf{Yes}$	$\mathbf{Yes}$	Yes
Individual-level controls	Yes	Yes	$\mathbf{Yes}$	Yes	$\mathbf{Yes}$	$\mathbf{Yes}$
Distributional norms	$ m Y_{es}$	Yes	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$
Additional controls	$N_{O}$	No	$\mathbf{Yes}$	No	$N_{O}$	Yes
Number of observations	539,035	401,933	401,933	539,035	401,933	401,933
p-value (F-statistic)	0.000	0.000	0.000	0.000	0.000	0.000
R-Squared	0.073	0.072	0.073	0.073	0.072	0.073
Notes: *, **, *** denotes statistical significance on the 10%, 5%, and 1% level, respectively. Standard errors are clustered by individual and are given in parentheses. Approximate elasticities of the dependent variable with respect to inequality perceptions are given in brackets. The additional individual-level controls are: highest completed education (in years), employment status (10 dummies), and personal income (measured as	tistical signif nd are given erceptions a (in years), e:	icance on th in parenthe re given in mployment :	le 10%, 5%, sees. Approx brackets. T status (10 du	and 1% level, cimate elasticit The additional mmies), and <sub>1</sub>	respectively. Sizes of the deperiod individual-leve	tandard errors ndent variable l controls are e (measured as

Table A.3: Reasons for getting ahead (non-meritocratic mechanisms only), extended set of individual-level controls